Author’s response to reviews

Title:A comparative effectiveness trial of postoperative management for lumbar spine surgery: changing behavior through physical therapy (CBPT) study protocol

Authors:

Kristin R Archer (kristin.archer@vanderbilt.edu)
Rogelio A Coronado (rogelio.coronado@vanderbilt.edu)
Christine M Haug (chrissy.m.haug@vanderbilt.edu)
Susan W Vanston (susan.w.vanston@vanderbilt.edu)
Clinton J Devin (clinton.j.devin@vanderbilt.edu)
Christopher J Fonnesbeck (chris.fonnesbeck@vanderbilt.edu)
Oran S Aaronson (oran.aaronson@vanderbilt.edu)
Joseph S Cheng (joseph.cheng@vanderbilt.edu)
Richard L Skolasky (rskolas@jhmi.edu)
Lee H Riley III (lriley@jhmi.edu)
Stephen T Wegener (swegener@jhmi.edu)

Version:2Date:23 September 2014

Author’s response to reviews:

We have removed the additional files as requested.